Author's response to reviews

Title: OK-DIABETES: Supported self-management for adults with type 2 diabetes and a learning disability: Study protocol a randomized controlled feasibility trial

Authors:

Rebecca EA Walwyn (R.E.A.Walwyn@leeds.ac.uk)
Amy M Russell (a.m.russell@leeds.ac.uk)
Louise D Bryant (l.d.bryant@leeds.ac.uk)
Amanda J Farrin (a.j.farrin@leeds.ac.uk)
Alex Wright-Hughes (a.wright-hughes@leeds.ac.uk)
Elizabeth H Graham (e.h.graham@leeds.ac.uk)
Claire T Hulme (c.t.hulme@leeds.ac.uk)
John L O'Dwyer (j.odwyer@leeds.ac.uk)
Gary Latchford (g.latchford@leeds.ac.uk)
Alison Stansfield (alisonstansfield@nhs.uk)
Dinesh Nagi (Dinesh.Nagi@midyorks.nhs.uk)
Ramzi Ajjan (r.ajjan@leeds.ac.uk)
Allan O House (a.o.house@leeds.ac.uk)

Version: 4 Date: 22 June 2015

Author's response to reviews: see over
Dear Sir/Madam,
I am delighted that you have decided to accept our paper provisionally. Enclosed is a revised manuscript, addressing the comments from the reviewers, entitled “OK-DIABETES: Supported self-management for adults with type 2 diabetes and a learning disability: Study protocol a randomized controlled feasibility trial”. Changes are tracked and below are point-by-point responses to each of the issues raised.

1. **p5 Eligibility criteria:** HbA1c "ideally within 6 weeks of randomization". Please add the maximum time range you set (if any)

   I have added “no maximum explicitly set” in brackets.

2. **Please add the "Trial Status" at the end of your manuscript.**

   This was already there p15-16 – although it was over two sides – so I have put it all on p16.

3. **It appears that 'LG' should be 'EG' in the Authors' Contributions section. Can you please correct this?**

   I have done this.

Yours sincerely,
Dr Rebecca Walwyn on behalf of the co-authors
Principal Statistician